Author's response to reviews

Title: Towards an understanding of barriers to condom use in rural Benin using the Health Belief Model: A cross sectional survey.

Authors:

Sennen H Hounton (hounton_sennen@yahoo.fr)
Helene Carabin (helene-carabin@ouhsc.edu)
Neil J Henderson (neil-henderson@ouhsc.edu)

Version: 8 Date: 10 January 2005

Author's response to reviews: see over
Re: Submission of Revisions on final version of Manuscript

Dear Editor, BMC Public Health,

I am writing to submit revisions on the manuscript untitled «Towards an Understanding of Barriers to Condom use in Rural Benin using the Health Belief Model: A cross sectional survey» following your 2005.

1) I have formatted the abstract to include Background, Methods, Results and Conclusions.

2) Column headings have been removed in Table 2

I thank you very much for your time and consideration to this revised manuscript and I’m looking forward in hearing from you any time soon.

Sincerely Yours,

Sennen Hounton, MD. MPH